Total score /100 pts. First team member Second team member

SE 2A04 Fall 2000 Lab Exercise 2 Marking Scheme

Instructor: William M. Farmer

1. Report

(a) Description and analysis of work /30 pts. (b) Source code and documentation

i. The team’s full program /15 pts. ii. The other team’s pair program /5 pts.

(c) Testing

i. The team’s check overlap program /5 pts. ii. The team’s compare trees program /5 pts. iii. The team’s full program /5 pts. iv. The team’s compare trees

with other team’s check overlap program /5 pts. (d) Log books

i. First team member’s /5 pts. ii. Second team members’s /5 pts.

2. Demonstration of the team’s full program /20 pts.